JAOC 20,6

56

Received 28 April 2023 Revised 27 October 2023 8 January 2024 Accepted 16 February 2024

Personality characteristics, preferences for rewards and the propensity to choose an auditing job

Iryna Alves

ISCAL (Lisbon Accounting and Business School), Lisbon Polytechnic Institute, Lisbon, Portugal and ISEG (Lisbon School of Economics and Management), Universidade de Lisboa and ADVANCE, CSG Research Centre, Lisbon, Portugal

Bruno Gregório ISEG (Lisbon School of Economics and Management), Universidade de Lisboa and ADVANCE, CSG Research Centre, Lisbon, Portugal and Europeia University, Lisbon, Portugal, and

Sofia M. Lourenço ISEG (Lisbon School of Economics and Management), Universidade de Lisboa and ADVANCE, CSG Research Centre, Lisbon, Portugal

Abstract

Purpose – This study investigates theoretical relationships among personality characteristics, preferences for different types of rewards and the propensity to choose a job in auditing by management-related higher education students. Specifically, the authors consider motivation, locus of control (internal and external) and self-efficacy (SE) as personality characteristics and financial, extrinsic, support and intrinsic as types of rewards.

 $\label{eq:Design/methodology/approach-Data were collected through a questionnaire targeted at management-related higher education students in Portugal. Partial least squares structural equation modelling was used to analyse the data.$

Findings – The full sample results show that different types of motivation, locus of control and SE are related to different reward preferences. The authors also find a positive association between a preference for extrinsic rewards and the propensity to choose a job in auditing. Moreover, when the authors consider the role of working experience in the model, the authors find that the reward preferences that drive the choice of an auditing job differ according to that experience.



Journal of Accounting & Organizational Change Vol. 20 No. 6, 2024 pp. 56-84 Emerald Publishing Limited 1832-5912 DOI 10.1108/JAOC 04-2023-0080 © Iryna Alves, Bruno Gregório and Sofia M. Lourenço. Published by Emerald Publishing Limited. This article is published under the Creative Commons Attribution (CC BY 4.0) licence. Anyone may reproduce, distribute, translate and create derivative works of this article (for both commercial and non-commercial purposes), subject to full attribution to the original publication and authors. The full terms of this licence may be seen at http://creativecommons.org/licences/by/4.0/legalcode

Authors thank the editor, Prof. Zahirul Hoque, two anonymous reviewers, Andson Braga de Aguiar, Aldónio Ferreira, Roland Speklé, Pinky Rusli, and participants at the XXII Grudis Conference and Doctoral Colloquium, 45th Annual Congress of the European Accounting Association, 12th Conference on Performance Measurement and Management Control, and 2024 Management Accounting Section Midyear Meeting of the American Accounting Association for helpful comments in earlier drafts. This paper was distinguished with the Outstanding Paper Award Professor Lúcia Lima Rodrigues at the XXII Grudis Conference and Doctoral Colloquium. Authors gratefully acknowledge financial support from FCT – Fundação para a Ciência e Tecnologia (Portugal), national funding through research grant UIDB/04521/ 2020. All errors remain our own.

Originality/value – This study enriches the literature by assessing preferences for different types of rewards, considering multiple personality characteristics and a comprehensive set of rewards. Furthermore, the authors identify the reward preferences that drive the choice of an auditing career. This knowledge empowers auditing firms to devise recruitment strategies that resonate with candidates' preferences, which boosts the capacity of these companies to attract new auditors.

Keywords Motivation, Auditing, Personality, Preferences, Incentives

Paper type Research paper

1. Introduction

Management-related students in higher education are often interested in choosing an auditing career (Espinosa-Pike *et al.*, 2021), usually working with clients from the private (instead of public) sector (Wen *et al.*, 2018). Nevertheless, new entrants in the profession are not enough to replace auditors who leave (Khavis and Krishnan, 2021), reflecting the high rates of turnover in the auditing industry (Nouri and Parker, 2020). Additionally, worldwide fraud scandals have, to a certain extent, eroded the image and social perception of the auditing profession (Bhaskar *et al.*, 2019; Herron and Gilbertson, 2004). The shortage in supply is even more acute because of an increasing demand for auditing services arising from environmental, social and governance reporting (Krasodomska *et al.*, 2021; Pearson, 2014).

In addressing this phenomenon, it is crucial to explore possible ways to attract auditor candidates by analysing factors that may affect the supply of students, i.e. those entering the job market who are intent on pursuing a career in auditing. This topic is of high interest to auditing researchers and other stakeholders in the industry because the shortage of talent in auditing is usually identified as one of the major problems that auditing companies face [Association of Chartered Certified Accountants (ACCA), 2020; Association of Chartered Certified Accountants and Accounting and Corporate Regulatory Authority (ACCA and ACRA), 2012].

Prior research on the choice of an auditing job has focused on personality traits (Fisher, 2001; Holt *et al.*, 2017) and on outcome expectations/career goals (Tetteh *et al.*, 2022). We bridge these two streams of research and develop a theoretical model in which personality characteristics are considered as determinants of reward preferences, which ultimately have an effect on the choice of an auditing job. Rewards are important when considering possible future jobs (Victor and Hoole, 2021), notably in auditing, because they have the power to motivate, influence and determine the performance, satisfaction and turnover in a given job. Rewards also affect job attractiveness to the extent to which they match one's preferences (Chiang and Birtch, 2007). Therefore, the growing interest in the drivers of self-reported preferences for rewards (Lourenço, 2020), such as employees' personality characteristics, is not surprising.

In fact, personality characteristics stand out as substantial predictors of one's preferences (Julian *et al.*, 2021), as they are at the root of human behaviour (Becker *et al.*, 2011). Personality has been increasingly incorporated in management accounting research, inasmuch as people do not make decisions only by applying cognitive elements (Abernethy and Wallis, 2019). That is, personality characteristics are known to have equally large effects on outcomes. Personality characteristics include, among others, motivation, *locus* of control and self-efficacy (SE). Studying these can contribute to a better understanding of reward systems' efficacy (Vandenberghe *et al.*, 2008). However, to the best of the authors' knowledge, no study attempts to establish a relationship between multiple personality characteristics and preferences for rewards, nor a relationship between preferences for rewards and the propensity to choose an auditing job (PAJ).

To examine these issues, we collected data from Portugal. The accounting and auditing industry in Portugal is valued at €1.2bn and is ranked 16th in Europe in 2023 (out of 27 European Union countries). In 2023, the accounting and auditing industry in Portugal is the 77th largest

JAOC 20.6

58

industry (of the 294 total industries that IBISWorld tracks) [1]. The industry's rank (77th) has remained stable since 2018 [2].

Although several accounting fraud scandals have affected the accounting and auditing professions all over the world, the professions in Portugal do not seem to have been significantly harmed (Leão and Gomes, 2022). The professional associations of accountants and auditors stand as the most prestigious and elitist corporatist organisations in Portugal (Rodrigues *et al.*, 2003). The Big 4 auditing companies have largely contributed to this image with their international prestige, technical knowledge, human resources and contact with international clients (de Almeida, 2012). The Portuguese enforcement mechanisms, also play an important role in raising professional standards of ethical behaviour for professional auditors (de Almeida, 2014), which contributes to the consistent development of this profession in terms of status and performance (Rodrigues *et al.*, 2003). According to Teixeira *et al.* (2015), auditing and accounting professions are common entry jobs for management-related students in Portugal. To attract new applicants to their firms, Big4 companies usually hold recruitment sessions in the large Portuguese universities with management-related programs, which encompass various courses in accounting, taxation and/or auditing. Therefore, in this context, management-related programs act as a pathway for future auditors.

We collect survey data from a sample of 652 management-related students from Portugal and use partial least squares structural equation modelling (PLS-SEM) to analyse them. Results from our full sample show that extrinsic motivation is positively associated with preferences for financial and extrinsic rewards and intrinsic motivation with support and intrinsic rewards. We also find that external *locus* of control (ELC) is positively associated with a preference for extrinsic rewards, and internal *locus* of control (ILC) relates positively to preferences for all types of rewards, except financial rewards. Additionally, SE has a positive association with preferences for financial, extrinsic rewards and the propensity to choose a job in auditing. Moreover, when we consider the role of working experience in our model, we find that the reward preferences that drive the choice of an auditing job differ according to that experience. Specifically, we find that for those with less or no working experience, a preference for extrinsic rewards is the only predictor of their choice of an auditing job, whereas for those with more working experience, a preference for financial rewards is the only significant predictor.

Our study contributes to recruitment and compensation research in auditing. Prior research highlights that the rewards used to direct and motivate audit staff (non-partners) are largely unexplored (Burrows and Black, 1998; Coram and Robinson, 2016; Fiolleau *et al.*, 2023). The few studies in this regard show that the importance placed on various rewards is a function of auditors' career stage (Lynn *et al.*, 1996; Miao *et al.*, 2009). Our study contributes to this stream of research by showing that rewards are also important to attract prospective job candidates and not only to motivate current auditors. Specifically, we identify the preferences for rewards that are related to students' willingness to become auditors and how these preferences change with their working experience.

The next sections are organised as follows. Section 2 reviews the literature and develops the research hypotheses. Section 3 presents the methodology and Section 4 presents the results. Finally, Section 5 concludes.

2. Literature review and hypotheses development

2.1 Choice of an auditing job

Previous studies examining the choice of an auditing job have predominantly centred on individual personality traits (Fisher, 2001; Holt *et al.*, 2017) and outcome expectations/career goals (Tetteh *et al.*, 2022). We join these streams in our research by proposing a more holistic

theoretical framework. In our framework, personality characteristics are posited to shape reward preferences, which in turn influence the decision to pursue an auditing job. The significance of rewards cannot be understated when evaluating prospective career paths (Victor and Hoole, 2021), especially in the auditing domain, because rewards play a pivotal role in driving motivation, shaping performance outcomes, enhancing job satisfaction and influencing turnover rates.

This is particularly important in the auditing industry, which has high rates of turnover (Nouri and Parker, 2020), especially in large firms (Khavis and Krishnan, 2021). Consequently, the imperative to attract new auditors has become even more pronounced. The literature provides several reasons for the high turnover rates, such as heavy workload and time pressure (Persellin *et al.*, 2019), unsatisfying work–life balance (Khavis and Krishnan, 2021) and scandals that erode the image of the auditing career, entailing distrust (Holtzblatt *et al.*, 2020). Rewards can serve as a pivotal mechanism to attract and retain talent, offsetting the departure of auditors from their careers.

2.2 Personality characteristics

Although several factors have been identified as determinants of career preference and selection, prior research highlights the role of personality in the choice of a given career (Barrick *et al.*, 2001). Each career requires an individual to hold certain personal qualities to perform the tasks involved and find enjoyment in doing so (Raveenther, 2017; Roberts and Robins, 2000). In the context of the auditing profession, Fisher (2001) suggests a connection between personality, notably Type A (characterized by competitiveness, drive, ambition, anger, hostility, irritability, impatience and time urgency) and the role of an auditor. Holt *et al.* (2017) use the Big Five Personality Model (which encompasses extraversion, agreeableness, openness, conscientiousness and neuroticism) to link accounting students' personalities to the choice of an auditing profession.

While the literature identifies personality types and traits that are related to the choice of auditing profession, the studies ignore other personality characteristics that shape the preferences for different incentives or rewards. Specifically, because personality characteristics will determine how a person adapts to their career (Holland, 1985), various job attributes such as working environment, monetary incentives, growth opportunities, social recognition and/or flexible working hours may be more or less appealing to persons with different characteristics. In this regard, motivation, *locus* of control and SE stand out as personality characteristics with the (theoretical) potential to influence reward preferences.

Motivation is a driver that leads human beings to do something, such as engaging in a certain action (Deci *et al.*, 2017). Motivation can vary not only in its strength – high or low – but also in its source – extrinsic or intrinsic (Deci and Ryan, 2000). Deci and Ryan (2000) conceptualize extrinsic motivation (ExtMot) as a driver of behaviour when individuals expect a separable outcome from the task/action. Conversely, individuals with intrinsic motivation (IntMot) engage in a certain task due to the joy and pleasure they gain from performing the task itself, not because they are looking for *ex post* outcomes (Gagné and Deci, 2005). Hence, IntMot comes from internal factors, whereas ExtMot is driven by external causes. These two concepts are therefore different in nature and, as such, demand different approaches from organisations when designing a reward system (Victor and Hoole, 2021).

Locus of control can be defined as an individual's beliefs about the influence s/he has over outcomes or what happens in his/her life (Rotter, 1966). The locus of control has two strands – external and internal. If someone is convinced that external causes play a dominant role in outcomes, leaving almost no space for personal influence, she/he has an ELC and she/he is known to be an external. Conversely, if an individual believes that his/her

own actions are the main determinants of a given outcome, she/he is considered to have an ILC and she/he is called an internal.

Finally, SE refers to the extent to which an individual thinks that she/he is able to undertake a task successfully (Bandura, 1978). Therefore, SE refers to the individual's perception of being competent (Lei *et al.*, 2021).

2.3 Preferences for rewards

IAOC

20.6

60

The aim of a reward system designed and used by an organisation is to attract, motivate and retain high-quality employees (Gomez-Mejia et al., 2004). Reward systems are designed according to what is considered to be employees' preferences (Lawler, 2000; Rynes et al., 2004). Preferences for rewards and their effects on job outcomes and job choices (Acheampong, 2021) are the subjects of ongoing debate in academia. For instance, contemporary research suggests that matching a stated preference for a reward with the actual receipt of that reward can either lead to a better job outcome (e.g. performance) or have no effect (Lourenco, 2020). A possible explanation for the no-effect outcome is that self-reported preferences may come with some pitfalls, namely, social desirability bias (Rynes et al., 2004) and a lay rationalism effect (Hsee et al., 2003). Social desirability bias occurs when individuals express their preferences according to social norms instead of their own genuine will. Lay rationalism refers to human beings' tendency to overrate "hard" characteristics, such as money, due to its tangible nature and fungibility, compared to "soft" characteristics, such as non-financial rewards that are difficult to quantify. These biases can also affect the relationships between preferences for rewards and job choices. Therefore, it is even more essential to consider various preferences for different rewards rather than framing the issue as a singular preference for rewards.

Aligned with the idea that there are multiple preferences for rewards, Bussin *et al.* (2019) suggest that reward preferences can differ based on demographic cohorts and describe rank preferences for types of rewards across the three generations currently active in the labour market. They find generation Y (those who are currently joining the workforce or have recently done so) to be the most demanding one, as they prioritize growth opportunities and learning experiences when choosing a job, placing these considerations immediately above compensation [3], [4]. Other studies suggest that generation Z individuals prioritize non-financial motivating rewards, such as career advancement opportunities for personal growth and development (lorgulescu, 2016), interesting and meaningful work (Schroth, 2019) and personal fulfilment (Grow and Yang, 2018). However, they also place significant value on tangible rewards such as monetary compensation (Goh and Lee, 2018), benefits at work such as health plans (Maloni *et al.*, 2019), family leave (Grow and Yang, 2018) and job security (Goh and Lee, 2018).

The multiplicity of rewards is also increasingly evident in the auditing industry. A recent study by Fiolleau *et al.* (2023) shows that beyond monetary rewards for non-partners, audit firms leverage non-monetary rewards such as gift card-based and points-based schemes that are redeemable for merchandise, etc. These are implemented for various purposes, from performance incentives to employee recognition and business development. Notably, while the audit industry has historically favoured penalty-based schemes, there has been growing advocacy for shifting towards more reward-centric approaches to inspire high-quality audits (Peecher *et al.*, 2013).

In response to this evolving landscape, our study draws upon Chiang and Birtch's (2007) reward typology. This typology ensures a comprehensive exploration of the diverse reward structures that auditing firms are now adopting. These authors classify rewards into four categories: one financial and three non-financial (extrinsic, support and intrinsic). The strength of using this classification resides in its ability to capture the vast diversity of non-financial incentives, which are conceptually distinct and serve different purposes. This framework enables

us to state our hypotheses with greater precision, considering the unique nature of each reward type and corresponding personality characteristics.

Financial rewards (FinRew) are cash-flow based rewards such as individual performance incentives, basic salary, overall benefits, team performance incentives, job security and an organisation incentive plan (Chiang and Birtch, 2007). Prior research shows that FinRew can increase the acceptance of difficult performance goals, leading to an enhancement of productivity (Jenkins *et al.*, 1998). Moreover, FinRew attracts above-average workers because only those who are likely to have good performance have a reason to self-select themselves into variable compensation agreements (Gerhart and Fang, 2014).

Extrinsic rewards (ExtRew) are non-financial (not cash-flow-based) and are related to causes external to the task in itself. Following Chiang and Birtch's (2007) classification, ExtRew includes relationships with co-workers, management style, authority/power, job pressure, job title/status, equity and team spirit. Past research has been especially focused on management style, power and status (Gkorezis and Petridou, 2012).

Support rewards (SuppRew) are non-financial (not cash-flow-based) rewards related to the working conditions that aim to provide joy from performing the task. Thus, they can be considered as more "internal" than the previous category. SuppRew may encompass organisational support, job location, alternative work arrangements, flexible benefits, work– life balance and working environment (Chiang and Birtch, 2007). Berenyi (2022) highlights the importance of non-financial rewards such as organisational support, work–life balance and the working environment for employees to fully perform their duties. Organisational support refers to someone feeling that his/her firm values employees' contributions and cares about their emotional well-being (Akingbola and van den Berg, 2019). In parallel, work–life balance offers more flexibility to workers to manage their lives, professionally and personally, which enhances job satisfaction (Carleton and Kelly, 2019). Finally, working environment refers to interpersonal relationships, that is, the extent to which ties between colleagues are guided by mutual help and fellowship, which can affect job satisfaction (Paramitha and Indarti, 2014).

Finally, intrinsic rewards (IntRew) are non-financial rewards related to the task in itself, i.e. internal factors. Examples of IntRew, as illustrated by Chiang and Birtch (2007), are: challenge, job variety, accomplishment, autonomy, responsibility, nature of work, opportunity to use skills/ability, learning opportunities and job satisfaction. Organisations that provide their employees with non-financial rewards such as autonomy, challenging tasks, opportunities to use skills and learning moments, are intrinsically rewarding their workforce (see Clay *et al.*, 2022).

2.4 Personality characteristics and preferences for rewards

Theoretically, employees prefer certain rewards over others, depending on their personality characteristics. Regarding motivation, we argue that different types of motivation will likely lead to different types of preferences for rewards. For example, FinRew and ExtRew are based on the instrumental value of doing something (Derfler-Rozin and Pitesa, 2021), namely, the possibility of being eligible to receive money or social recognition (Stajkovic and Luthans, 2001). Hence, ExtMot should be a driver of these preferences, as this type of motivation refers to external motives to engage in a certain task. FinRew and ExtRew encourage people to undertake a given task by providing them reasons to do so beyond the task itself. Hence, we argue that ExtMot should be positively related to a preference for financial rewards (PreFinRew) and a preference for extrinsic rewards (PreExtRew) because these rewards have an external nature. Therefore, our hypotheses are the following:

JAOC 20.6

62

- H1a. ExtMot is positively associated with PreFinRew.
- H1b. ExtMot is positively associated with PreExtRew.

Conversely, IntMot is about the joy and pleasure of doing a task just for the sake of doing so. This joy may arise because the person finds the job to be interesting or because it represents what she/he stands for (Gagné and Deci, 2005). Both SuppRew and IntRew aim to provide conditions that foster the joy of undertaking a given job. SuppRew aims to provide good working conditions so employees can perform their work at their maximum, thereby establishing an emotional tie between the employee and the employer. Hence, SuppRew target internal factors, enhancing a sense of relatedness with the company (Akingbola and van den Berg, 2019). IntRew are means to design more interesting jobs that favour the inclusion of, for example, learning opportunities and challenging tasks (Victor and Hoole, 2021). Hence, we argue that intrinsic motivation should be positively related to a preference for support rewards (PreSuppRew) and a preference for intrinsic rewards (PreIntRew) as they have an internal nature. Thus, our hypotheses are the following:

- H1c. IntMot is positively associated with PreSuppRew.
- H1d. IntMot is positively associated with PreIntRew.

Locus of control (LC) is also a predictor of preferences for rewards (Heywood *et al.*, 2017) because it refers to the perception that individuals have over outcomes. Specifically, externals usually present low levels of self-esteem (Ng et al., 2006) and high levels of neuroticism (or lack of emotional stability) (Bono and Judge, 2003), which leads lack to а of confidence. Therefore, externals push themselves away from situations that, in spite of being positive in nature, entail a payoff. This is because externals feel they will not be able to meet others' expectations, i.e. they shield themselves from the possibility of being, in their minds, a potential disappointment (Majerczyk et al., 2020) by rejecting the situation from the outset to avoid feeling anxious (Spector, 1982). Rewards in general (financial, extrinsic, support and intrinsic) are assigned to employees to motivate them, so rewards aim to reach a positive outcome. However, rewards also convey an idea of employees' obligation to deliver results (Baker et al., 1988). This idea of delivering better results – especially for those who are more self-doubting, such as externals – triggers anxiety. Therefore, we expect that externals shield themselves from these feelings, leading them to discard being rewarded. Thus, our hypothesis is the following:

H2. ELC is negatively associated with preferences for rewards.

In opposition to externals, internals are usually more optimistic individuals (Ratnawati *et al.*, 2021). This optimism may arise from the belief they have in being in charge of their lives (i.e. in control of what they get), which leads them to think they will achieve desirable outcomes. Moreover, internals also seek help to attain the goals they have set for themselves (Singh *et al.*, 2020), which increases their probability of achievement. According to Ng *et al.* (2006), in a work setting, internals search for jobs/tasks that have high motivating characteristics, such as rewards. In fact, this idea is consistent with Malik *et al.*'s (2015) argument that internals do not see incentives ("external interventions") as a pressure condition but rather as an aid to reach the goals of both themselves and their organisation. Therefore, even if rewards, as stated above, may be seen as a means that organisations have to demand a higher performance (that ultimately is difficult to attain), we argue that internals will see this as an opportunity to defy themselves instead of being afraid not to match others' expectations (unlike externals). Hence, our hypothesis is the following:

H3. ILC is positively associated with preferences for rewards.

Because SE refers to the perceptions that individuals have about their probability of success when executing a task, SE is also an important determinant of preferences for rewards. Individuals with more confidence in their personal ability (high SE) perceive greater expectations of their actions and are likely to be more attracted to compensation systems that link their individual behaviour to rewards (Cable and Judge, 1994). In fact, SE is a significant predictor of performance (Kader, 2022) both at work and in academic settings. Individuals with high SE seek positive reinforcements, others' approval and the feeling of thriving at work (Ashraf *et al.*, 2014). Thus, we argue that individuals with high SE – who like to demonstrate their potential – perceive all rewards as an opportunity to show their abilities and also be acknowledged for their performance. For instance, the prospect of being rewarded with money (financial reward), power (extrinsic reward), a healthy working environment (support reward), or learning opportunities (intrinsic reward) is perceived by individuals with high SE as an opportunity to use their technical, leadership, interpersonal and cognitive skills, respectively. Therefore, we hypothesize the following:

H4. SE is positively associated with preferences for rewards.

2.5 Preferences for rewards and propensity to choose an auditing job

Preferences for rewards can have a strong influence on job preferences and career decisions (Cable and Judge, 1994). Reward systems in companies may act as signalling devices for potential candidates, affecting job and organizational attractiveness by providing information about less visible organizational attributes (Gerhart and Milkovich, 1992; Rynes and Miller, 1983). Prospective job candidates make search and application decisions based on their perception of the alignment between their own preferences and organizational characteristics (Turban and Keon, 1993). Central to this argument is the nature of the organization's human resource system (Bretz and Judge, 1994), with a particular emphasis on the rewards available.

Therefore, in our theoretical model, we conceptualize how preferences for rewards drive the choice of an auditing job. Specifically, we consider the PAJ as the extent to which an individual is willing to apply for a job in auditing based on its attractiveness to him/her according to the reward preferences she/he has. In this regard, we expect that PAJ should be aligned with how well auditing companies offer rewards (financial, extrinsic, support and intrinsic) that match the preferences of the potential candidates. For example, prior research suggests that a factor that could affect students' propensity to choose accounting-related jobs is the salary scale (Jackling and Keneley, 2009; Awadallah and Elgharbawy, 2021). However, other studies (see Frecka *et al.*, 2022) show that auditing firms do not provide above-average starting salaries and that they even underperform similar job options (such as tax or corporate accounting). Moreover, this gap persists over the years (Hoopes *et al.*, 2018). An auditing job is also known for its long hours of work, which makes the financial package even less competitive in relative terms. This means that if students place a great emphasis on financial rewards, i.e. have a strong PreFinRew, they will be less interested in an auditing-related career. Thus, we hypothesize the following:

H5a. PreFinRew is negatively associated with PAJ.

Similarly, other types of extrinsic preferences, such as power, status or a certain type of leadership style, provided by auditing companies are expected to influence PAJ. Specifically, auditing companies are known by their fast promotion ladder (Pruijssers *et al.*, 2020) and also by their open and communicative leadership style (Duh *et al.*, 2020; Nekhili *et al.*, 2021).

Propensity to choose an auditing job

63

IAOC Moreover, auditors are viewed as professionals with a positive image regarding their reliability, timeliness, courtesy and respect (Tetteh et al., 2022). As such, students who value these extrinsic factors, i.e. have strong PreExtRew, will be more interested in an auditrelated career choice. Hence, we hypothesize the following:

H5b. PreExtRew is positively associated with PAJ.

20.6

64

Intrinsic factors are also important in the choice of a specific job (Awadallah and Elgharbawy, 2021). In this regard, SuppRew, which is related to perceived organisational support and a healthy working environment, is a special challenge in the auditing profession. Work overload is a structural feature in auditing, as documented by the literature over many years (Hegazy et al., 2023; Persellin et al., 2019). The lack of work-life balance is also often reported by those who have chosen to become auditors (Khavis and Krishnan, 2021), because it is a consequence of work overload. Work overload and work-life (dis)balance are also a concern for auditing regulatory bodies, which see them as threats to the attractiveness of an auditing job [Association of Chartered Certified Accountants and International Federation of Accountants (ACCA and IFAC), 2021]. Furthermore, the lack of auditors' determination to whistleblow regarding unethical behaviours (Erkmen et al., 2014; Wainberg and Perreault, 2016) suggests a lack of organisational support from auditing firms. Therefore, students who place a great emphasis on organisational support and a healthy working environment, i.e. have a strong PreSuppRew, will be less likely to pursue an auditing job. Thus, we hypothesize the following:

H5c. PreSuppRew is negatively associated with PAJ.

Finally, intrinsic rewards such as challenging and variable tasks, autonomy, increased responsibility and learning opportunities, are also important for the choice of an auditing profession. Wen et al. (2018) argue that students who believe that accounting-related professions provide a wider variety of professional experiences are more positive about an accounting-related career than others. In fact, individuals who work at auditing firms report constant apprenticeship moments (Kusaila, 2019), classifying auditing as a challenging profession not only because of its content but also because of the variety of industries requiring audits (Low, 2004). Similarly, Jackson et al. (2023) suggest that the dynamic nature of the accounting-related profession is one of the most attractive factors for students. Moreover, professional independence in the workplace positively influences students' intention to choose an accounting-related job (Wen et al., 2015). Concurrently, those who choose to become auditors expect to have great responsibility as well, because investors and, more broadly, stakeholders rely on auditors' opinions to make decisions (see Chen et al., 2021). Therefore, students who place a great emphasis on intrinsic rewards, i.e. have a strong PreIntRew, will be more likely to pursue an auditing job. Thus, we hypothesize the following:

H5d. PreIntRew is positively associated with PAJ.

Figure 1 illustrates our conceptual model.

3. Research method

3.1 Data gathering technique and survey design

Due to the nature of the variables in our model, which are not available via secondary sources, we use survey-collected primary data. The survey, hosted on Qualtrics, was distributed to undergraduate and graduate students in management-related areas at a large university and a large polytechnic institute in Portugal via a link shared with potential respondents [5]. Additionally, other contacts with students and professors in other schools in Portugal were



Source: Author's own creation

made to increase the number of responses. Linkedin was also used to reach more potential respondents (only within the network of students enrolled in management-related courses) due to its network visibility. To minimize social desirability bias, which is a common issue in behavioural studies, we clarified that neither wrong nor right answers exist, we ensured respondents of the confidentiality of their answers, and we provided the possibility of choosing the option "do not know/do not answer" in all questions.

To increase the response rate, we indicated that it took only 10 min to complete the questionnaire, and we used both reminders and financial incentives (participants who completed the questionnaire were eligible to win one of three vouchers of \notin 50 each). Additionally, we offered participants the possibility of receiving the conclusions of our study as an additional (non-financial) incentive.

We collected 1,077 responses over a two-month period between May and July 2022, but only 652 were usable, i.e. had less than 15% missing values (Hair *et al.*, 2017) [6]. We are not able to compute a response rate because participants could share the survey link with their peers. Additionally, we cannot test for potential (non-)response bias, as we are not able to differentiate early from late respondents as participants were invited to participate in the survey on different days. Nevertheless, we address the concern of common method bias (CMB). CMB can arise because we use the same measurement instrument to collect data for both our endogenous and exogenous variables (Speklé and Widener, 2018). Therefore, we perform a full collinearity assessment, following Kock's (2015) procedure. Because all values are below 3.3, CMB is not a matter of concern in our model (Kock, 2015). Moreover, we also perform Harman's one factor test to assess CMB (Podsakoff *et al.*, 2003). The results suggest, once again, that CMB is not a concern because the average variance extracted (AVE) of a single latent variable was 20.8%, which is lower than 50% (Kock, 2021).

The analyses of our sample show that female participants account for 67% of the respondents. The age of our respondents varies from 18 to 65 years, with a mean of 23 and a mode of 19.52% of the participants report having working experience, but about half of them were in the labour market for less than one year.

3.2 Variables measurement

The questions and items were adapted from previously validated scales, and we pre-tested our survey to ensure its clarity and syntax (Speklé and Widener, 2018). Items' loadings are presented in the Appendix. Table 1 presents descriptive statistics for all constructs.

Extrinsic motivation (ExtMot) and intrinsic motivation (IntMot) were measured using scales conceived by Gagné et al. (2010). The IntMot scale has three indicators. For ExtMot, we added two items from van der Kolk et al. (2019), thereby making a scale comprising five items, Responses were made on a seven-point Likert scale, for which 1 = Not at all and 7 = Very strongly.

ELC and ILC were measured with Lumpkin's (1985) scale, which has six items (three for each construct). Respondents were asked about their level of agreement with each indicator, responding on a five-point Likert scale ranging from 1 = Strongly disagree to 5 = Strongly agree.

SE was measured with a scale initially proposed by Chen et al. (2001) and modified by Imperial College London (2019). This scale has six items responded to on a five-point Likert scale (1 = Not at all confident, 5 = Extremely confident).

Preference for financial rewards (PreFinRew), Preference for extrinsic rewards (PreExtRew), Preference for subport rewards (PreSuppRew) and Preference for intrinsic rewards (PreIntRew) were assessed with Chiang and Birtch's (2007) scales. We asked respondents to express their level of appreciation for each reward (item) in a future job on a five-point Likert scale ranging from 1 (I do not give importance) to 5 (I give great importance). This scale has 28 items divided into four types of preferences.

PAJ was measured with questions and items adapted from Bartlett et al. (2017). Respondents were asked about their level of agreement with each item and responded on a seven-point Likert scale (1 = Strongly disagree to 7 = Strongly agree).

3.3 Planned analyses

For the PAJ scale, we start by performing an exploratory factor analysis (EFA) using principal component method. We identify a single latent variable and use it in further analyses.

Variable	N	P25	Mean	Median	P75	St dev
ExtMot	645	4.40	5.05	5.20	5.80	1.09
IntMot	639	4.33	4.96	5.00	5.67	1.18
ELC	639	2.00	2.63	2.67	3.00	0.68
ILC	648	3.33	3.57	3.67	4.00	0.60
SE	648	2.50	2.97	3.00	3.50	0.85
PreFinRew	641	3.67	4.00	4.00	4.33	0.56
PreExtRew	637	3.29	3.71	3.71	4.07	0.55
PreSuppRew	648	3.83	4.15	4.17	4.50	0.55
PreIntRew	643	3.78	4.13	4.11	4.55	0.52
PAI	597	3.40	4.42	4.60	5.60	1.40

Table 1. Descriptive statistics for the measurement

variables

Source: Authors' own creation

IAOC

20.6

Next, we test our theoretical model using PLS-SEM with SmartPLS 4.0 (Ringle *et al.*, 2022). We use PLS-SEM rather than a covariance-based (CB-SEM) technique for model estimation due to the nature of our data (Hair *et al.*, 2017). Indeed, CB-SEM requires the distribution of variables to follow a bell-shaped curve (Hair *et al.*, 2017). That condition is seldomly observable, especially in behavioural research that uses psychometric variables (Goodhue *et al.*, 2012). Conversely, PLS-SEM is a distribution-free data analysis technique that is more suitable for our data.

PLS-SEM model assessment is a two-step procedure. In the first stage, we compute a measurement model to infer constructs' reliability and validity (convergent and discriminant). In addition, all constructs in our measurement model are reflective because latent variables give rise to manifest variables and there are no second-order constructs (Bedford and Speklé, 2018). In the second stage, we use a structural model to test the research hypotheses, after checking that collinearity was not an issue (all variance inflation factors were below 5).

4. Results and discussion

4.1 Results of exploratory factor analysis

The Kaiser–Meyer–Olkin (KMO = 0.872) and the Bartlett's test of sphericity (p < 0.01) support the EFA for the variable PAJ, and the Kaiser's criterion allowed extracting one factor with an eigenvalue of 3.64 (the factor explained 73% of total variance and Cronbach's alpha = 0.9).

4.2 Reflective measurement model evaluation

To examine the measurement model, we start by assessing indicators' reliability using outer loadings. Loadings above 0.708 should be kept, and those below 0.4 must be excluded (Hair *et al.*, 2017). For those loadings that range between these limits, we delete them (each one at a time) only if, by doing so, it would improve the composite reliability (CR) and/or AVE beyond threshold values – which are 0.7 and 0.5, respectively (Hair *et al.*, 2017).

Furthermore, regarding internal consistency, we focused on analysing Cronbach's alpha (hereinafter, alpha) and CR, because true reliability of measures is lower bounded by the former and upper bounded by the latter (Hair *et al.*, 2017). Alpha's acceptable threshold value is 0.5-0.6 (Nunnally, 1978). Analysing the alphas and CR of each construct, we conclude that all latent variables pass the alpha criterion, with the exception of two – ELC and ILC – even though both of them meet CR's standard. Because the alpha measure is truly conservative in its computation formula, it could provide underestimated reliability (Hair *et al.*, 2017). We opted to keep these constructs in our model so that all hypotheses could be tested, but in a robustness test, we dropped them to infer the quality of our inferences without these constructs.

To examine convergent validity, AVE was assessed. Each latent variable attained the 0.5 threshold, which means that every construct explains at least half of its items' variance (Hair *et al.*, 2017). In addition, all indicators are statistically significant (all of them at the 1% significance level) with a bootstrapping procedure using 5,000 subsamples. In conclusion, there is no concern about convergent validity in our model.

To analyse discriminant validity, we used Fornell and Larcker's (1981) yardstick, which demands that each construct share more variance with its items than with other latent variables' indicators. Table 2 presents the results of these procedures with the square root of AVE in italic diagonal. All inter-construct correlations are lower than items' square root variance explained by their construct for any variable, thus indicating that each latent variable is unique and, therefore, discriminant validity exists.

JAOC 20.6		1	2	3	4	5	6	7	8	9	10	Alpha	CR	AVE
_ • , •	1. ExtMot	0.762										0.831	0.873	0.580
	2. IntMot	0.249	0.842									0.796	0.880	0.710
68	3. ELC	-0.009	-0.157	0.807								0.466	0.789	0.652
	4. ILC	0.143	0.190	-0.279	0.806							0.482	0.785	0.649
	5. SE	0.081	0.259	-0.251	0.187	0.868						0.935	0.948	0.754
	4. PreFinRew	0.268	0.046	-0.027	0.106	0.130	0.747					0.733	0.834	0.558
	7. PreExtRew	0.401	0.134	0.035	0.158	0.221	0.510	0.805				0.712	0.843	0.648
Table 2. Convergent and	8. PreSuppRew	0.184	0.215	-0.076	0.117	0.053	0.499	0.373	0.746			0.631	0.784	0.557
	9. PreIntRew	0.120	0.381	-0.160	0.225	0.365	0.301	0.319	0.470	0.709		0.836	0.876	0.503
	10. PAJ	0.248	0.202	-0.096	0.141	0.097	0.129	0.165	0.093	0.091	0.839	0.889	0.921	0.704
discriminant validity evidence	Source: Autho	ors' own	creation											

4.3 Structural model evaluation

Table 3 shows the empirical results of our conceptual model (base model).

The base model column in Table 3 shows that ExtMot is positively and significantly related to both PreFinRew and PreExtRew, thus supporting *H1a* and *H1b*. These findings are consistent with Stajkovic and Luthans' (2001) arguments, conveying the idea that

Expected sign	Independent variable	Dependent variable	Base model Coeff.	Model 1 Coeff.	Model 2 Coeff.
+	ExtMot	PreFinRew	0.252***	0.260***	0.252***
+		PreExtRew	0.371***	0.386***	0.371***
+	IntMot	PreSuppRew	0.202***	0.217***	0.202***
+		PreIntRew	0.288***	0.307***	0.288***
_	ELC	PreFinRew	0.016		0.016
_		PreExtRew	0.116***		0.116***
_		PreSuppRew	-0.029		-0.029
_		PreIntRew	-0.016		-0.016
+	ILC	PreFinRew	0.056		0.056
+		PreExtRew	0.100**		0.100**
+		PreSuppRew	0.075*		0.075*
+		PreIntRew	0.116***		0.116***
+	SE	PreFinRew	0.103**	0.107**	0.103**
+		PreExtRew	0.202***	0.188***	0.202***
+		PreSuppRew	-0.020	-0.000	-0.020
+		PreIntRew	0.265***	0.285***	0.265***
_	PreFinRew	PAJ	0.052	0.050	0.055
+	PreExtRew	PAJ	0.126***	0.127**	0.130***
_	PreSuppRew	PAJ	0.004	0.007	-0.014
+	PreIntRew	PAJ	0.033	0.033	0.041
	RA	PAJ			0.091**
Model fit		Ν	652	652	652
		SRMR	0.068	0.070	0.066
		rms Theta	0.124	0.123	0.124

Table 3.Note: *, ** and *** indicate significant path coefficient at *p*-value level of 10, 5 and 1%, respectively (two-tailed)Base model and
robustness checksSource: Authors' own creation

individuals whose motivation is based on the instrumental value of doing something prefer rewards that target external factors, such as FinRew and ExtRew.

We also find support for *H1c* and *H1d*, because IntMot is positively and significantly associated with both PreSuppRew and PreIntRew. These findings are also aligned with previous research (Gagné and Deci, 2005), suggesting that individuals who engage in tasks either because these tasks are interesting by themselves or because they represent what the individual stands for will look for rewards that enhance this feeling of inner motivation, namely, IntRew and SuppRew.

Interestingly, *H2*, predicting a negative relationship between ELC and preferences for rewards, is not supported. Specifically, ELC is not significantly related to PreFinRew, PreSuppRew and PreIntRew. However, ELC is positively and significantly related to PreExtRew, which is contrary to our hypothesis of a negative relationship. A possible explanation for the positive relationship is that externals tend to feel more unattended and socially excluded (Ye and Lin, 2015), and when ExtRew are provided, they may release a sense of social recognition given by others (Stajkovic and Luthans, 2001), thereby reducing feelings of social exclusion or unattendance. This may lead externals to state a preference for ExtRew. Alternatively, because externals usually experience powerless feelings (Desai *et al.*, 2018), they may attempt to offset these feelings by seeking power in the workplace. In fact, ExtRew provides power, among other things.

H3, referring to a positive relationship between ILC and preferences for all rewards, is partially validated. Specifically, ILC is positively associated with PreExtRew, PreSupRew and PreIntRew. These findings are consistent with prior literature (Malik *et al.*, 2015; Ng *et al.*, 2006), suggesting that internals see rewards as a way to reach their professional desires. Nonetheless, ILC is not significantly related to PreFinRew, which means that the previous argument does not hold for financial incentives. A possible explanation for this could be social desirability bias (Rynes *et al.*, 2004). Internals are individuals who like to project a good image of themselves (Valentine *et al.*, 2019). In western societies, stating an open preference for money may lead to an association with greed (Zeelenberg *et al.*, 2020), which has a negative connotation. Thus, internals may refrain from openly stating a preference for FinRew, leading to a non-significant coefficient between ILC and PreFinRew.

H4, referring to a positive relationship between SE and preferences for all rewards, is also partially validated. Specifically, SE is positively related to PreFinRew, PreExtRew and PreIntRew. These results are aligned with our argument that high-SE individuals like to display their potential and perceive rewards as an opportunity to exhibit their abilities, which leads to a high preference for rewards. Our results show, however, that this reasoning does not apply to SuppRew, as the relationship is not significant. A possible explanation for this is that individuals who perceive themselves as being effective are usually high achievers. According to Rayburn and Rayburn (1996), high achievers have a Type A personality, which means they are chronically competitive. Their competitiveness, in turn, leads them to be emotionally detached from others (Jia *et al.*, 2022). Because SuppRew aims to build an emotional tie between the employee and the employer, high-SE individuals' emotional detachment may explain the absence of a preference for SuppRew.

Finally, we find a positive association only between PreExtRew and PAJ, thus validating *H5b*, but we fail to find statistically significant relationships for other preferences. Therefore, our data do not support *H5a*, *H5c* and *H5d*. The positive and significant relationship between PreExtRew and PAJ advances prior qualitative studies suggesting that social recognition is the main driver to becoming an auditor amongst management-related students (Bekoe *et al.*, 2018; Tetteh *et al.*, 2022), and ExtRew

IAOC provides, among others, social recognition. Figure 2 shows the final structural model with significant relationships. 20.6

4.4 Robustness tests

To analyse the strength of our findings, we perform several robustness tests. In Model 1, Table 3, we drop both ELC and ILC because both constructs were below the lower bound reliability range threshold (alpha < 0.5). Model 1 shows that our results are similar to the base model: all statistical inferences are unaffected as all path coefficients' significance levels remain similar.

In Model 2, Table 3, we include risk-aversion (hereinafter, RA) as a control variable related to the propensity to choose a job in auditing [7]. RA can be defined as the extent to which a person likes to mitigate risks. In fact, an auditing job is a work of great responsibility (Chen et al., 2021), so planning plays a key role and aims to mitigate risks. Therefore, it is likely that those who are risk-averse tend to like auditing, thus establishing a positive association between RA and PAJ. Indeed, we obtain a positive and statistically significant association (coeff. = 0.091, p < 0.05). More importantly, all results remain gualitatively unchanged (with similar significance levels) when compared to those of the base model.

We perform a third robustness test (untabulated) by running our model on a subsample of only management-related students. Even though our questionnaire was distributed mainly to management-related students, we have a few observations that come from other areas. Because students from non-management-related courses (e.g. law, data management and quantitative methods) can have different motivations and preferences regarding their career choice, we run our main model without this group of



Figure 2. Final structural model with significant relationships



70

students (50 observations). We find that all statistical inferences are qualitatively unchanged, as all path coefficients' significance levels are similar to those in the main model.

4.5 Additional analyses

Source: Authors' own creation

Because about half of our sample includes students with working experience, which can influence students' career decisions to become auditors (Navallas *et al.*, 2022) and their reward preferences, we perform some additional analyses.

First, we include a moderator variable for working experience (measured as the number of months of working experience) in the main model (linking it to each type of reward preferences–PAJ relationship) and find that all statistical inferences remain as all path coefficients' significance levels and their signals are similar to those of the main model. We also find that working experience positively and significantly moderates the relationship between PreFinRew and PAJ (coeff. = 0.145, p < 0.05).

To further investigate the effect of working experience, we split our sample into two groups – one without working experience and the other with working experience – and ran our main model in these subsamples (applying a bootstrapping procedure with 5,000 subsamples). Table 4 shows the results for these subsamples.

We find that for the sample of students without working experience, the relationship between PreExtRew and PAJ is still the only statistically significant one (coeff. = 0.156, p < 0.05), and all the other preferences for rewards are not related to PAJ. However, for the

Independent variable	Dependent variable	Sample without working experience Coeff.	Sample with working experience Coeff.	Difference in coefficients	Sample with low working experience Coeff.	Sample with high working experience Coeff.	Difference in coefficients
ExtMot	PreFinRew	0.239***	0.244***	-0.005	0.202***	0.338***	-0.136
	PreExtRew	0.446***	0.316***	0.130**	0.367***	0.346***	0.021
IntMot	PreSuppRew	0.187***	0.199***	-0.013	0.157***	0.297**	-0.139
	PreIntRew	0.347***	0.231***	0.116	0.288***	0.229**	0.059
ELC	PreFinRew	-0.084	0.087	-0.170*	0.017	-0.057	0.074
	PreExtRew	0.052	0.177***	-0.125	0.119***	0.057	0.063
	PreSuppRew	-0.016	-0.028	0.013	0.008	-0.188	0.196
	PreIntRew	-0.054	-0.013	-0.041	-0.027	-0.026	-0.001
ILC	PreFinRew	0.024	0.118**	-0.093	0.077	0.118	-0.041
	PreExtRew	0.018	0.161***	-0.143*	0.116***	0.146	-0.030
	PreSuppRew	0.085	0.113	-0.028	0.139***	0.104	0.035
	PreIntRew	0.091	0.125**	-0.034	0.105**	0.283	-0.178
SE	PreFinRew	0.114	0.077	0.037	0.118**	0.068	0.051
	PreExtRew	0.207***	0.182***	0.025	0.217***	0.116	0.100
	PreSuppRew	-0.014	-0.017	0.002	0.015	-0.097	0.112
	PreIntRew	0.236***	0.288***	-0.053	0.282***	0.194*	0.088
PreFinRew	PAJ	0.043	0.158*	-0.115	0.018	0.340**	-0.322*
PreExtRew	PAJ	0.156**	0.087	0.069	0.164***	-0.116	0.280*
PreSuppRew	PAJ	-0.012	-0.056	0.044	-0.010	-0.042	0.031
PreIntRew	PAJ	0.064	0.017	0.047	0.023	0.110	-0.087
	Ν	312	340		543	109	
Note: *, ** tailed)	and *** indic	ate significant	t path coefficie	ent at <i>p</i> -value	e level of 10, 5	and 1%, respec	tively (two-

Table 4.

Main model per subsamples of working experience JAOC 20.6

72

sample of students with working experience, we find that the relationship between PreFinRew and PAJ becomes significant (coeff. = 0.158, p < 0.1), while the PreExtRew–PAJ relationship becomes nonsignificant (p = 0.267) [8]. However, the test for the difference in coefficients between the two groups is not statistically significant for these relationships. The remainder of the coefficients for these subsamples is in general similar to those reported for the base model (Table 3), except for the *locus* of control [9].

Nevertheless, business practice indicates that after three years of professional experience, professionals are positioned to pursue roles with greater responsibility, with the corresponding financial incentives. Following this rationale, we split our sample into two groups – one without or with low working experience (i.e. fewer than three years) and the other with high working experience (i.e. 3 or more years) – and ran our main model in these subsamples (applying a bootstrapping procedure with 5,000 subsamples). The last three columns of Table 4 show the results for these subsamples. We find that for the sample of students with less or no working experience, the relationship between PreExtRew and PAJ is the only statistically significant (coeff. = 0.164, *p* < 0.01). For the sample of students with high working experience, we find that the relationship between PreFinRew and PAJ becomes significant (coeff. = 0.340, *p* < 0.05), while the PreExtRew–PAJ relationship becomes nonsignificant. Moreover, the test for the difference in coefficients between the two groups is statistically significant. In this analysis, the remainder of the coefficients for these subsamples are in general similar to those reported for the base model, except for the *locus* of control.

These findings regarding differences in the relationships between preferences for rewards and PAJ based on working experience provide important insights. They show that prospective candidates may be looking for different things in the auditing profession. Those without or with low working experience look for extrinsic rewards such as power, status and leadership style. Conversely, those with high working experience may enter the auditing profession as seniors or as managers. For them, extrinsic rewards may not be as important because they may already be established players in the job market. For them, financial rewards are valuable as they capitalize on their prior working experience. Additionally, because participants with working experience are also older (correlation coeff. = 0.77, p < 0.01), they may already have financial responsibilities and therefore place greater value on stability and pay in an auditing career. In fact, untabulated analyses performed on two subsamples (using a median split based on the age of the participants) yielded results similar to those derived from the subsamples based on working experience, especially concerning the relationships between preference for different types of rewards and PAJ [10].

These results advance prior research showing that the importance that individuals place on various rewards differs according to their career stage (Lynn *et al.*, 1996; Miao *et al.*, 2009) or demographic cohort (Bussin *et al.*, 2019). Specifically, employees in the exploration stage (usually between the ages of 20 and 30, during which an individual is primarily concerned with finding an occupation in which she/he can succeed and grow as an individual) prefer support and peer acceptance; in the establishment stage (usually between the ages of 30 and 45, during which a conscious commitment is made to a particular occupational field, and effort is expended to stabilize oneself) individuals are motivated by promotion and challenge; employees in the maintenance stage (starting generally around the late 30s to mid-40s, a time of holding one's own and maintaining what has already been achieved) are less competitive and value security (as well as salary, bonus and fringe benefits); while those in the disengagement stage (the final critical adjustment for employees is the transition from working to retirement) are moving into retirement and are likely to focus on issues related to their life after retiring (Lynn *et al.*, 1996; Miao *et al.*, 2009). Our results, using a recent cohort of applicants, unlike those used in prior studies, suggest that for those without or with low working experience (prior to or at the beginning of their exploration stage), extrinsic rewards, such as power, status and leadership style, are important in the decision to apply to an auditing job. These participants have not yet started (or just started) their careers, and, therefore, they may consider an auditing job as a means to gain recognition in the job market, which is key if they aim to move to another job in the future. Our study shows that those with high working experience (in the exploration stage or more advanced) stress more financial rewards, such as basic salary, performance incentives, job security and overall financial benefits, in their decision to seek an auditing job. This may reflect their more demanding financial responsibilities and the more competitive compensation packages offered by auditing companies for those who already have some working experience.

Regarding demographic cohorts, various studies explore the reward preferences of different generations. For example, prior studies found that individuals from generation Z are motivated by rewards of a non-financial nature, such as career advancement, opportunities for learning, growth and development (lorgulescu, 2016), interesting and meaningful work (Schroth, 2019), personal fulfilment (Grow and Yang, 2018) and good working relationships with their co-workers (Goh and Lee, 2018; Grow and Yang, 2018). Nevertheless, in some studies, pay (Goh and Lee, 2018), benefits at work such as health plans (Maloni et al., 2019) and family leave (Grow and Yang, 2018), as well as job security (Goh and Lee, 2018), also rank high in generation Z reward preferences. Conversely, millennials place a strong emphasis on financial performance-based rewards and job security (Magni and Manzoni, 2020), among other factors. Money could serve as a motivational tool, satisfying millennials' need for achievement or a sense of entitlement (Ng et al., 2010). Another factor influencing millennials' preference for financial-based rewards is their life stage, which often involves significant financial commitments such as purchasing homes and starting families (Barhate and Dirani, 2022). Our untabulated results support these arguments. Specifically, using Seemiller and Grace's (2016) classification, we split our sample into two groups: generation Z (i.e. up to 28 years old) and millennials (i.e. 28 years old or more) and ran our main model in these subsamples. The results suggest that, in their decision to seek an auditing job, millennials value financial rewards more than generation Z (the difference in coefficients between the two groups is statistically significant, coeff. = -0.374; p < 0.10). Therefore, our findings substantiate the claim that for older individuals (millennials), financial rewards are important in their decision to apply for an auditing job. This is consistent with prior research showing that millennials have a pronounced preference for financial rewards, which may reflect their more demanding financial responsibilities. Our study adds to this prior research by showing that this preference for financial rewards is a driver of their choice to pursue an auditing career.

5. Conclusions

Auditing companies struggle to find new auditors to replace those who leave (Khavis and Krishnan, 2021), with high turnover rates being a structural feature in the auditing industry (Nouri and Parker, 2020). Our study addresses this issue by providing insights regarding the choice of an auditing job by management-related students, considering different personality characteristics and preferences for different types of rewards.

We find a positive association between ExtMot and both PreFinRew and PreExtRew. We also find a positive association between IntMot and PreSuppRew, as well as between IntMot and PreIntRew. Such findings reinforce previous research positing that people whose energy is propelled by external drivers prefer money and extrinsic incentives (Stajkovic and

Luthans, 2001), whereas individuals whose motivation is driven by internal sources call for rewards that target internal factors (Gagné & Deci, 2005). Regarding LC, ELC is positively associated with PreExtRew, suggesting that externals try to offset their common feelings of social exclusion/unattendance (Ye and Lin, 2015) with social recognition provided by ExtRew assignment, ILC is positively associated with preferences for all rewards, with the exception of PreFinRew. This nonsignificant relationship might be due to social desirability bias, because previous research documents a highly significant association between social desirability and ILC (Valentine et al., 2019). We also find that SE is positively associated with preferences for all rewards, with the exception of PreSuppRew. A possible explanation for this is that high-SE individuals are high achievers with Type A personalities (Rayburn and Rayburn, 1996), which means that they are overcompetitive and tend to be emotionally detached (Jia *et al.*, 2022). Because SuppRew aims to build an emotional tie between the employee and the employer (Chiang and Birtch, 2007), high-SE individuals' emotional detachment leads to a nonsignificant relationship between SE and PreSuppRew. Finally, we find that only PreExtRew is positively associated with students' propensity to pursue a job in auditing.

Moreover, when we consider the role of working experience (age) in our model, we find that the reward preferences that drive the choice of an auditing job differ according to that experience (age). Specifically, we find that for those without or with less working experience (younger), PreExtRew is the only predictor of the choice of an auditing job, whereas for those with more working experience (older), PreFinRew is the significant predictor. These are important findings, as they show that prospective candidates may be looking for different things in the auditing profession. Those without working experience (younger) look for extrinsic rewards such as power, status and leadership style. Conversely, those with working experience (older) may enter the auditing profession as seniors or as managers. For them, power, status and leadership style may not be as important because they may already have an established name in the job market. For them, financial rewards are important as they capitalize on their prior working experience, and as they grow older, they likely also have greater financial responsibilities (Barhate and Dirani, 2022), such as purchasing homes and starting families.

Our findings provide a more detailed picture of the drivers of the choice of a career in auditing. Specifically, students not only expect certain outcomes, such as money and social prestige (Tetteh *et al.*, 2022), but the outcomes they expect differ according to their working experience (age). Additionally, these results advance the auditing literature by suggesting that the importance placed on various rewards is not only a function of auditors' career stage (Lynn *et al.*, 1996; Miao *et al.*, 2009) or age but also influences the decision of prospective candidates (i.e. management-related students) to choose an auditing job.

Our study has practical implications as our findings provide valuable insights for the recruitment strategies of auditing firms, which struggle with high turnover rates (Nouri and Parker, 2020). For example, by highlighting the rewards most important to the choice of an auditing job – non-financial extrinsic and financial – our study can help auditing companies to develop effective recruitment strategies to attract new applicants (Espinosa-Pike *et al.*, 2021), considering their previous working experience. However, past research shows that auditors with high intrinsic motivation are more likely to prioritize the profession itself, are less inclined to engage in unethical behaviour, and are more prone to whistleblowing (Erkmen *et al.*, 2014; Wainberg and Perreault, 2016). Therefore, audit firms should also consider the risks of an excessive focus on financial and extrinsic rewards, as this may discourage prospective candidates with strong intrinsic motivation.

IAOC

20.6

As with any research, this study is not exempt from limitations, and its findings should therefore be interpreted with caution. First, our analysis relies on cross-sectional data, so it is not possible to empirically establish causal relationships (Lourenço, 2019). We can only argue causal relationships at a theoretical level. Second, our study uses a self-reported measure of the PAJ and not an actual choice of applying to such a job (or actually entering the profession). Even though this measure shows high reliability and validity, it ultimately may not be translated into an actual choice.

Our study provides several avenues for future research. Although our setting is in the auditing profession, we believe that our findings can be generalizable to other settings. The relationships between personal characteristics and reward preferences reflect innate traits and therefore could be applicable to a variety of different settings. The relationships between reward preferences and the choice of an auditing job could be generalizable to professions with work environments comparable to the auditing sector, emphasizing extrinsic and financial rewards. Future research could explore these theoretical relationships in different industries and investigate how the specific context affects the relationships between preferences for rewards and career choices. This line of inquiry could provide further insights for the recruitment and compensation literature. Longitudinal research can help document causal relationships from an empirical perspective, which in our research are only posited at a theoretical level. Furthermore, other moderating effects could also provide a more in-depth understanding of how preferences for rewards affect the choice of an auditing job. For example, future research might investigate the role of academic background in the relationships we document. It would also be interesting to investigate whether our findings are specific to auditing within for-profit organizations. Auditing in the public sector can have different drivers, as auditors play a role in the democratic accountability of these organizations (Hay and Cordery, 2021). Similarly, auditing charities or non-governmental organizations may be seen as a different job, as auditors are gatekeepers for the different stakeholders of these organizations. Country comparative analysis would also be an interesting extension of our work because cultural differences may influence the perception of an auditing job by management-related students and the drivers of that choice. Finally, subsequent studies could delve deeper into an individual's intrinsic motivation as opposed to the extrinsic rewards commonly offered by auditing firms, as these can influence the profiles of potential hires and impact their propensity to speak up and share information within an auditing firm (Kadous and Zhou, 2019).

Notes

- IBISWorld is a global industry research firm that provides detailed reports and analysis on various industries used by banks and financial firms, academic institutions, government organizations, consulting firms, and corporations.
- Data retrieved from www.ibisworld.com/portugal/industry-statistics/accounting-auditing/3880/ on 21 August 2023.
- 3. Generation Y, also known as millennials, typically includes individuals born between 1981 and 1995, whereas Generation Z, or post-millennials, covers those born after 1995 (e.g. Seemiller and Grace, 2016). However, Bussin *et al.* (2019) classify generation Y as those born between 1981 and 2000, thereby merging what many recent studies treat as two distinct groups: generation Y and Z.
- 4. Generation Z closely resembles the demographic of the students we surveyed in this study, as we discuss in Section 3.
- 5. We targeted management-related students because they are the most likely to become auditors (Espinosa-Pike *et al.*, 2021).

JAOC 20,6	6. To avoid duplicate responses, we limited responses to one questionnaire per IP address and we requested an email address.
76	7. RA is a four-item variable that was adapted from Payan <i>et al.</i> (2012), whose responses were given on a five-point Likert scale (1 = Strongly disagree, 5 = Strongly agree). After dropping two indicators due to low outer loadings, we managed to reach all thresholds: $alpha = 0.71$; $CR = 0.87$; $AVE = 0.77$. We also checked for discriminant validity and collinearity, and no problems were detected.
	8. We conduct similar analyses excluding respondents aged 24 years and older (who are above percentile 75) and observe that the results are inferentially identical. Therefore, we find no evidence that outliers in terms of age are driving our results.
	9 The non-results for the locus of control relationships may be the product of the smaller sample

- 9. The non-results for the locus of control relationships may be the product of the smaller sample size (the number of observations in the main model is roughly twice the number of observations in each subsample), the weaker construct validity of the locus of control variables, and/or it may suggest that locus of control is not an important determinant of reward preferences for participants without working experience. Because these participants may not have experienced the different rewards, they may not associate them with internal or external causes.
- 10. Specifically, for younger participants, the relationship between PreExtRew and PAJ is the only statistically significant one; conversely, for older participants the relationship between PreFinRew and PAJ is the only statistically significant one.

References

- Abernethy, M.A. and Wallis, M.S. (2019), "Critique on the 'manager effects' research and implications for management accounting research", *Journal of Management Accounting Research*, Vol. 31 No. 1, pp. 3-40.
- Acheampong, N.A.A. (2021), "Reward preferences of the youngest generation: attracting, recruiting, and retaining generation Z into public sector organizations", *Compensation and Benefits Review*, Vol. 53 No. 2, pp. 75-97.
- Akingbola, K. and van den Berg, H.A. (2019), "Antecedents, consequences, and context of employee engagement in nonprofit organizations", *Review of Public Personnel Administration*, Vol. 39 No. 1, pp. 46-74.
- Ashraf, N., Bandiera, O. and Jack, B.K. (2014), "No margin, no mission? A field experiment on incentives for public service delivery", *Journal of Public Economics*, Vol. 120, pp. 1-17.
- Association of Chartered Certified Accountants and Accounting and Corporate Regulatory Authority (ACCA and ACRA) (2012), *Talent Attraction and Retention in Larger Accounting Firms* [Brochure], ACCA, ACRA, Singapore.
- Association of Chartered Certified Accountants and International Federation of Accountants (ACCA and IFAC) (2021), "Groundbreakers: Gen Z and the future of accountancy [brochure]", London, available at: www.ifac.org/knowledge-gateway/developing-accountancy-profession/publications/groundbreakers-gen-z-and-future-accountancy (accessed 7 October 2022).
- Association of Chartered Certified Accountants (ACCA) (2020), "Covid-19 global survey: inside business impacts and responses", London, UK. [Brochure], available at: www.accaglobal.com/my/en/ professional-insights/global-economics/Covid-19_A-Global-Survey.html (accessed 7 October 2022).
- Awadallah, E. and Elgharbawy, A. (2021), "Utilizing the theory of reasoned action in understanding students' choice in selecting accounting as major", *Accounting Education*, Vol. 30 No. 1, pp. 86-106.
- Baker, G.P., Jensen, M.C. and Murphy, K.J. (1988), "Compensation and incentives: Practice vs. Theory", *The Journal of Finance*, Vol. 43 No. 3, pp. 593-616.

Bandura, A. (1978), "Self-efficacy: toward a unifying theory of behavioral change", Advances in Behaviour Research and Therapy, Vol. 1 No. 4, pp. 139-161.

- Barhate, B. and Dirani, K.M. (2022), "Career aspirations of generation Z: a systematic literature review", European Journal of Training and Development, Vol. 46 Nos 1/2, pp. 139-157.
- Barrick, M.R., Mount, M.K. and Judge, T.A. (2001), "Personality and performance at the beginning of the new millennium: What do we know and where do we go next?", *International Journal of Selection and Assessment*, Vol. 9 Nos 1/2, pp. 9-30.
- Bartlett, G.D., Kremin, J., Saunders, K.K. and Wood, D.A. (2017), "Factors influencing recruitment of non-accounting business professionals into internal auditing", *Behavioral Research in Accounting*, Vol. 29 No. 1, pp. 119-130.
- Becker, A., Deckers, T., Dohmen, T., Falk, A. and Kosse, F. (2011), "The relationship between economic preferences and psychological personality measures", *Annual Review of Economics*, Vol. 4 No. 1, pp. 453-478.
- Bedford, D.S. and Speklé, R.F. (2018), "Construct validity in survey-based management accounting and control research", *Journal of Management Accounting Research*, Vol. 30 No. 2, pp. 23-58.
- Bekoe, R.A., Owusu, G.M.Y., Ofori, C.G., Essel-Anderson, A. and Welbeck, E.E. (2018), "Attitudes towards accounting and intention to major in accounting: a logistic regression analysis", *Journal* of Accounting in Emerging Economies, Vol. 8 No. 4, pp. 459-475.
- Berenyi, L. (2022), "The right to a healthy working environment in the light of flexible working conditions", *Revue Européenne du Droit Social*, Vol. 56 No. 3, pp. 70-79.
- Bhaskar, K., Flower, J. and Sellers, R. (2019), Disruption in the Audit Market: The Future of the Big Four, Routledge.
- Bono, J.E. and Judge, T.A. (2003), "Core Self-Evaluations: a review of the trait and its role in job satisfaction and job performance", *European Journal of Personality*, Vol. 17 No. 1_suppl, pp. 5-18.
- Bretz, R.D., Jr and Judge, T.A. (1994), "The role of human resource systems in job applicant decision processes", *Journal of Management*, Vol. 20 No. 3, pp. 531-551.
- Burrows, G. and Black, C. (1998), "Profit sharing in Australian big 6 accounting firms: an exploratory study", Accounting, Organizations and Society, Vol. 23 Nos 5/6, pp. 517-530.
- Bussin, M.H.R., Mohamed-Padayachee, K. and Serumaga-Zake, P. (2019), "A total rewards framework for the attraction of generation y employees born 1981–2000 in South Africa", SA Journal of Human Resource Management, Vol. 17, pp. 1-15.
- Cable, D.M. and Judge, T.A. (1994), "Pay preferences and job search decisions: a person-organization fit perspective", *Personnel Psychology*, Vol. 47 No. 2, pp. 317-348.
- Carleton, C. and Kelly, M. (2019), "Alternative work arrangements and job satisfaction", Atlantic Economic Journal, Vol. 47 No. 3, pp. 293-309.
- Chen, G., Gully, S.M. and Eden, D. (2001), "Validation of a new general Self-Efficacy scale", *Organizational Research Methods*, Vol. 4 No. 1, pp. 62-83.
- Chen, K., Chen, T.Y., Han, W. and Yuan, H. (2021), "Auditors under fire: the association between audit errors and the career setbacks of individual auditors", *Journal of Accounting Research*, Vol. 60 No. 3, pp. 1-48.
- Chiang, F.F.T. and Birtch, T. (2007), "The transferability of management practices: Examining crossnational differences in reward preferences", *Human Relations*, Vol. 60 No. 9, pp. 1293-1330.
- Clay, G., Mlynski, C., Korb, F.M., Goschke, T. and Job, V. (2022), "Rewarding cognitive effort increases the intrinsic value of mental labor", *Proceedings of the National Academy of Sciences*, Vol. 119 No. 5, pp. 1-8.
- Coram, P. and Robinson, M. (2016), "Professionalism and performance incentives in accounting firms", Accounting Horizons, Vol. 31 No. 1, pp. 103-123.
- de Almeida, B.J.M. (2014), "Ethics of business organizations: the basis for their sustainability", Journal of Management and Sustainability, Vol. 4 No. 2, pp. 46-69.

Propensity to choose an auditing job

77

de Almeida, B.M. (2012),	"Statutory auditing i	n Portugal:	historical	and market	evolution",	Tékhne
Vol. 10 No. 1, pp. 54	-72.					

- Deci, E.L. and Ryan, R.M. (2000), "Intrinsic and extrinsic motivations: classic definitions and new directions", *Contemporary Educational Psychology*, Vol. 25 No. 1, pp. 54-67.
- Deci, E.L., Olafsen, A.H. and Ryan, R.M. (2017), "Self-Determination theory in work organizations: the state of a science", Annual Review of Organizational Psychology and Organizational Behavior, Vol. 4 No. 1, pp. 19-43.
- Derfler-Rozin, R. and Pitesa, M. (2021), "Motivation purity bias: expression of extrinsic motivation undermines perceived intrinsic motivation and engenders bias in selection decisions", *Academy* of *Management Journal*, Vol. 63 No. 6, pp. 1840-1864.
- Desai, N., Dalal, S. and Rawal, S. (2018), "The effects of volunteerism on self-deception and locus of control", voluntas: International Journal of Voluntary and Nonprofit Organizations, Vol. 29 No. 1, pp. 83-92.
- Duh, R., Knechel, W. and Lin, C. (2020), "The effects of audit firms' knowledge sharing on audit quality and efficiency", auditing: A Journal of Practice and Theory, Vol. 39 No. 2, pp. 51-79.
- Erkmen, T., Çalışkan, A.Ö. and Esen, E. (2014), "An empirical research about whistleblowing behavior in accounting context", *Journal of Accounting and Organizational Change*, Vol. 10 No. 2, pp. 229-243.
- Espinosa-Pike, M., Aldazabal, E. and Barrainkua, I. (2021), "Undergraduate business students' perception of auditing: impact of proximity and knowledge on auditors' stereotype", *Managerial Auditing Journal*, Vol. 36 No. 5, pp. 699-723.
- Fiolleau, K., MacTavish, C. and Obendorf, G. (2023), "Tangible rewards for more than just productivity: examining Canadian public accounting firms' rewards programs", *Accounting Perspectives*, Vol. 22 No. 3, pp. 315-340.
- Fisher, R.T. (2001), "Role stress, type a behavior pattern, and external auditor job performance and satisfaction", *Behavioral Research in Accounting*, Vol. 13 No. 1, pp. 143-170.
- Fornell, C. and Larcker, D. (1981), "Evaluating structural equation models with unobservable variables and measurement error", *Journal of Marketing Research*, Vol. 18 No. 1, pp. 39-50.
- Frecka, T.J., Mittelstaedt, H.F. and Stevens, J.S. (2022), "Career paths and compensation for accounting graduates", Accounting Horizons, Vol. 36 No. 1, pp. 77-109.
- Gagné, M. and Deci, E.L. (2005), "Self-determination theory and work motivation", *Journal of Organizational Behavior*, Vol. 26 No. 4, pp. 331-362.
- Gagné, M., Forest, J., Gilbert, M.H., Aubé, C., Morin, E. and Malorni, A. (2010), "The motivation at work scale: validation evidence in two languages", *Educational and Psychological Measurement*, Vol. 70 No. 4, pp. 628-646.
- Gerhart, B. and Fang, M. (2014), "Pay for (individual) performance: Issues, claims, evidence and the role of sorting effects", *Human Resource Management Review*, Vol. 24 No. 1, pp. 41-52.
- Gerhart, B. and Milkovich, G.T. (1992), "Employee compensation: Research and practice", in M.D. Dunnette and L.M. Hough (Eds), *Handbook of Industrial and Organizational Psychology*, Vol. 3, pp. 481-569. Consulting Psychologists Press, Inc, Palo Alto, CA.
- Gkorezis, P. and Petridou, E. (2012), "The effect of extrinsic rewards on public and private sector employees' psychological empowerment: a comparative approach", *The International Journal of Human Resource Management*, Vol. 23 No. 17, pp. 3596-3612.
- Goh, E. and Lee, C. (2018), "A workforce to be reckoned with: the emerging pivotal generation Z hospitality workforce", *International Journal of Hospitality Management*, Vol. 73, pp. 20-28.
- Gomez-Mejia, L.R., Balkin, D.B. and Cardy, R.L. (2004), *Managing Human Resources*, 4th ed., Prentice Hall, Upper Saddle River, NJ.
- Goodhue, D.L., Lewis, W. and Thompson, R. (2012), "Does PLS have advantages for small sample size or Non-Normal data?", *MIS Quarterly*, Vol. 36 No. 3, pp. 981-1001.

JAOC 20.6

Grow, J.M. and Yang, S. (2018), "Generation-Z enters the advertising workplace: expectations through a gendered lens", <i>Journal of Advertising Education</i> , Vol. 22 No. 1, pp. 7-22.	Propensity to
Hair, J., Hult, G., Ringle, C. and Sarstedt, M. (2017), A Primer on Partial Least Squares Structural	auditing job

- Hair, J., Hult, G., Ringle, C. and Sarstedt, M. (2017), A Primer on Partial Least Squares Structural Equation Modeling (PLS-SEM), 2nd ed., Sage, Thousand oaks CA.
- Hay, D. and Cordery, C.J. (2021), "Evidence about the value of financial statement audit in the public sector", Public Money and Management, Vol. 41 No. 4, pp. 304-314.
- Hegazy, M., El-Deeb, M.S., Hamdy, H.I. and Halim, Y.T. (2023), "Effects of organizational climate, role clarity, turnover intention, and workplace burnout on audit quality and performance", Journal of Accounting and Organizational Change, Vol. 19 No. 5, pp. 765–789.
- Herron, T.L. and Gilbertson, D.L. (2004), "Ethical principles vs. ethical rules: the moderating effect of moral development on audit independence judgments", Business Ethics Quarterly, Vol. 14 No. 3, pp. 499-523.
- Heywood, J.S., Jirjahn, U. and Struewing, C. (2017), "Locus of control and performance appraisal", Journal of Economic Behavior and Organization, Vol. 142, pp. 205-225.
- Holland, I.L. (1985), Making Vocational Choices: A Theory of Careers, 2nd ed., PrenticeHall, Englewood Cliffs, NI.
- Holt, T., Burke-Smalley, L.A. and Jones, C. (2017), "An empirical investigation of student career interests in auditing using the big five model of personality", Advances in Accounting Education: Teaching and Curriculum Innovations, Emerald Publishing Limited, Bingley, pp. 1-31.
- Holtzblatt, M.A., Foltin, C. and Tschakert, N. (2020), "Learning from ethical violations in public accounting: a South African audit scandal and a firm's transformation", Issues in Accounting Education, Vol. 35 No. 2, pp. 37-63.
- Hoopes, J.L., Merkley, K.J., Pacelli, J. and Schroeder, J.H. (2018), "Audit personnel salaries and audit quality", Review of Accounting Studies, Vol. 23 No. 3, pp. 1096-1136.
- Hsee, C.K., Zhang, J., Yu, F. and Xi, Y. (2003), "Lay rationalism and inconsistency between predicted experience and decision", Journal of Behavioral Decision Making, Vol. 16 No. 4, pp. 257-272.
- Imperial College London (2019), "General self-efficacy scale", available at: www.imperial.ac.uk/ education-research/evaluation/what-can-i-evaluate/self-efficacy/tools-for-assessing-self-efficacy/ general-self-efficacy-scale/ (accessed 26 February 2022).
- Iorgulescu, M.C. (2016), "Generation Z and its perception of work", Cross-Cultural Management Journal, Vol. 18 No. 1, pp. 47-54.
- Jackling, B. and Keneley, M. (2009), "Influences on the supply of accounting graduates in Australia: a focus on international students". Accounting and Finance, Vol. 49 No. 1, pp. 141-159.
- Jackson, D., Richardson, J., Michelson, G. and Munir, R. (2023), "Perceptions and experiences of career success among aspiring and early career accountants and the role of organisational support", Accounting and Finance, Vol. 63 No. 1, pp. 229-245.
- Jenkins, G.D., Mitra, A., Gupta, N. and Shaw, J.D. (1998), "Are financial incentives related to performance? A Meta-Analytic review of empirical research", Journal of Applied Psychology, Vol. 83 No. 5, pp. 777-787.
- Jia, R., Bahoo, R., Cai, Z. and Jahan, M. (2022), "The Hexaco personality traits of higher achievers at the university level", Frontiers in Psychology, Vol. 13, pp. 1-10.
- Julian, A.M., Wijngaard, O. and de Vries, R.E. (2021), "Compensation preferences: the role of personality and values", Frontiers in Psychology, Vol. 12, pp. 1-9.
- Kader, A.A. (2022), "Locus of control, self-efficacy, and student performance in an introductory economics course", International Review of Economics Education, Vol. 39, p. 100234.
- Kadous, K. and Zhou, Y. (2019), "How does intrinsic motivation improve auditor judgment in complex audit tasks?", Contemporary Accounting Research, Vol. 36 No. 1, pp. 108-131.

79

JAOC	Khavis, J. and Krishnan, J. (2021), "Employee satisfaction and work-life balance in accounting firms and audit quality", <i>auditing: A Journal of Practice and Theory</i> , Vol. 40 No. 2, pp. 161-192.						
20,0	Kock, N. (2015), "Common method bias in PLS-SEM: a full collinearity assessment approach", International Journal of e-Collaboration, Vol. 11 No. 4, pp. 1-10.						
	Kock, N. (2021), "Harman's single factor test in PLS-SEM: checking for common method bias", <i>Data Analysis Perspectives Journal</i> , Vol. 2 No. 2, pp. 1-6.						
80	Krasodomska, J., Simnett, R. and Street, D.L. (2021), "Extended external reporting assurance: current practices and challenges", <i>Journal of International Financial Management and Accounting</i> , Vol. 32 No. 1, pp. 104-142.						
	Kusaila, M.M. (2019), "Impact of informal learning in the accounting profession", <i>Journal of Workplace Learning</i> , Vol. 31 No. 7, pp. 429-441.						
	Lawler, E.E. (2000), <i>Rewarding Excellence: Pay Strategies for the New Economy</i> , Jossey-Bass Inc, San Francisco, CA.						
	Leão, F. and Gomes, D. (2022), "The stereotype of accountants: using a personality approach to assess the perspectives of laypeople", <i>Accounting, Auditing and Accountability Journal</i> , Vol. 35 No. 9, pp. 234-271.						
	Lei, W., Zhang, H., Deng, W., Wang, H., Shao, F. and Hu, W. (2021), "Academic self-efficacy and test anxiety in high school students: a conditional process model of academic buoyancy and peer support", <i>School Psychology International</i> , Vol. 42 No. 6, pp. 616-637.						
	Lourenço, S.M. (2019), "Field experiments in managerial accounting research", <i>Foundations and Trends® in Accounting</i> , Vol. 14 No. 1, pp. 1-72.						
	Lourenço, S.M. (2020), "Do self-reported motivators really motivate higher performance?", <i>Management Accounting Research</i> , Vol. 47, pp. 1-13.						
	Lumpkin, J.R. (1985), "Validity of a brief locus of control scale for survey research", <i>Psychological Reports</i> , Vol. 57 No. 2, pp. 655-659.						
	Lynn, S.A., Cao, L.T. and Horn, B.C. (1996), "The influence of career stage on the work attitudes of male and female accounting professionals", <i>Journal of Organizational Behavior</i> , Vol. 17 No. 2, pp. 135-149.						
	Magni, F. and Manzoni, B. (2020), "Generational differences in workers' expectations: Millennials want more of the same things", <i>European Management Review</i> , Vol. 17 No. 4, pp. 901-914.						
	Majerczyk, M., Owens, J. and Waddoups, N. (2020), "Contract framing, expected disappointment, and effort: the moderating role of external locus of control", <i>Journal of Management Accounting Research</i> , Vol. 32 No. 2, pp. 183-200.						
	Malik, M.A.R., Butt, A.N. and Choi, J.N. (2015), "Rewards and employee creative performance: Moderating effects of creative self-efficacy, reward importance, and locus of control", <i>Journal of</i> <i>Organizational Behavior</i> , Vol. 36 No. 1, pp. 59-74.						
	Maloni, M., Hiatt, M.S. and Campbell, S. (2019), "Understanding the work values of gen Z business students", <i>The International Journal of Management Education</i> , Vol. 17 No. 3, pp. 1-13.						
	Miao, C.F., Lund, D.J. and Evans, K.R. (2009), "Reexamining the influence of career stages on salesperson motivation: a cognitive and affective perspective", <i>Journal of Personal Selling and</i> <i>Sales Management</i> , Vol. 29 No. 3, pp. 243-255.						
	Navallas, B., Campo, C.D. and Camacho-Miñano, M.D.M. (2022), "Professional contacts and the decision to become an auditor. An analysis using LinkedIn", <i>Accounting Education</i> , Vol. 33 No. 1, pp. 1-19.						
	Nekhili, M., Javed, F. and Nagati, H. (2021), "Audit partner gender, leadership and ethics: the case of earnings management", <i>Journal of Business Ethics</i> , Vol. 177 No. 2, pp. 233-260.						
	Ng, E.S., Schweitzer, L. and Lyons, S.T. (2010), "New generation, great expectations: a field study of the millennial generation", <i>Journal of Business and Psychology</i> , Vol. 25 No. 2, pp. 281-292.						

Ng, T.W.H., Sorensen, K.L. and Eby, L.T. (2006), "Locus of control at work: a meta-analysis", <i>Journal of Organizational Behavior</i> , Vol. 27 No. 8, pp. 1057-1087.	Propensity to
Nouri, H. and Parker, R.J. (2020), "Turnover in public accounting firms: a literature review", <i>Managerial</i> <i>Auditing Journal</i> , Vol. 35 No. 2, pp. 294-321.	auditing job

- Nunnally, J. (1978), Psychometric Theory, 2nd ed., McGraw-Hill, New York, NY.
- Paramitha, A. and Indarti, N. (2014), "Impact of the environment support on creativity: assessing the mediating role of intrinsic motivation", *Procedia - Social and Behavioral Sciences*, Vol. 115, pp. 102-114.
- Payan, J.M., Svensson, G. and Hpogevold, N.M. (2012), "The effect of attributes of study abroad and risk aversion on the future likelihood to study abroad: a study of U.S. and Norwegian undergraduate marketing students", *Journal for Advancement of Marketing Education*, Vol. 20 No. 3, pp. 70-83.
- Pearson, D. (2014), "Significant reforms in public sector audit-staying relevant in times of change and challenge", *Journal of Accounting and Organizational Change*, Vol. 10 No. 1, pp. 150-161.
- Peecher, M.E., Solomon, I. and Trotman, K.T. (2013), "An accountability framework for financial statement auditors and related research questions", *Accounting, Organizations and Society*, Vol. 38 No. 8, pp. 596-620.
- Persellin, J., Schmidt, J., Vandervelde, S. and Wilkins, M. (2019), "Auditor perceptions of audit workloads, audit quality, and job satisfaction", *Accounting Horizons*, Vol. 33 No. 4, pp. 95-117.
- Podsakoff, P.M., MacKenzie, S.B., Lee, J.Y. and Podsakoff, N.P. (2003), "Common method biases in behavioral research: a critical review of the literature and recommended remedies", *Journal of Applied Psychology*, Vol. 88 No. 5, pp. 879-903.
- Pruijssers, J.L., Heugens, P.P. and van Oosterhout, J. (2020), "Winning at a losing game? Side-Effects of perceived tournament promotion incentives in audit firms", *Journal of Business Ethics*, Vol. 162 No. 1, pp. 149-167.
- Ratnawati, V., Setyosari, P., Ramli, M. and Atmoko, A. (2021), "Development of academic optimism model in learning for junior high school students", *European Journal of Educational Research*, Vol. 10 No. 4, pp. 1741-1753.
- Raveenther, A. (2017), "The relationship between career choice and environment, opportunity and personality", *International Journal of Scientific and Engineering Research*, Vol. 8 No. 3, pp. 63-67.
- Rayburn, J. and Rayburn, L. (1996), "Relationship between Machiavellianism and type a personality and ethical-orientation", *Journal of Business Ethics*, Vol. 15 No. 11, pp. 1209-1219.
- Ringle, C.M., Wende, S. and Becker, J.-M. (2022), SmartPLS 4, SmartPLS, Oststeinbek, available at: www.smartpls.com
- Roberts, B.W. and Robins, R.W. (2000), "Broad dispositions, broad aspirations: the intersection of personality traits and major life goals", *Personality and Social Psychology Bulletin*, Vol. 26 No. 10, pp. 1284-1296.
- Rodrigues, L.L., Gomes, D. and Craig, R. (2003), "Corporatism, liberalism and the accounting profession in Portugal since 1755", Accounting Historians Journal, Vol. 30 No. 1, pp. 95-128.
- Rotter, J.B. (1966), "Generalized expectancies for internal versus external control of reinforcement", *Psychological Monographs: General and Applied*, Vol. 80 No. 1, pp. 1-28.
- Rynes, S.L. and Miller, H.E. (1983), "Recruiter and job influences on candidates for employment", *Journal of Applied Psychology*, Vol. 68 No. 1, pp. 147-154.
- Rynes, S.L., Gerhart, B. and Minette, K.A. (2004), "The importance of pay in employee motivation: discrepancies between what people say and what they do", *Human Resource Management*, Vol. 43 No. 4, pp. 381-394.
- Schroth, H. (2019), "Are you ready for gen Z in the workplace?", California Management Review, Vol. 61 No. 3, pp. 5-18.
- Seemiller, C. and Grace, M. (2016), Generation Z Goes to College, Josey-Bass, San Francisco.

81

JAOC 20,6	Singh, L.B., Kumar, A. and Srivastava, S. (2020), "Academic burnout and student engagement: a moderated mediation model of internal locus of control and loneliness", <i>Journal of International</i> <i>Education in Business</i> , Vol. 14 No. 2, pp. 219-239.
	Spector, P.E. (1982), "Behavior in organisations as a function of employee's locus of control", <i>Psychological Bulletin</i> , Vol. 91 No. 3, pp. 482-497.
82	Speklé, R.F. and Widener, S.K. (2018), "Challenging issues in survey research: discussion and suggestions", <i>Journal of Management Accounting Research</i> , Vol. 30 No. 2, pp. 3-21.
	Stajkovic, A.D. and Luthans, F. (2001), "Differential effects of incentive motivators on work performance", <i>Academy of Management Journal</i> , Vol. 44 No. 3, pp. 580-590.
	Teixeira, C., Gomes, D. and Borges, J. (2015), "Introductory accounting students' motives, expectations and preparedness for higher education: some Portuguese evidence", Accounting Education, Vol. 24 No. 2, pp. 123-145.
	Tetteh, L.A., Agyenim-Boateng, C., Kwarteng, A., Muda, P. and Sunu, P. (2022), "Utilizing the social cognitive career theory in understanding students' choice in selecting auditing as a career: evidence from Ghana", <i>Journal of Applied Accounting Research</i> , Vol. 23 No. 3, pp. 715-737.
	Turban, D.B. and Keon, T.L. (1993), "Organizational attractiveness: an interactionist perspective", Journal of Applied Psychology, Vol. 78 No. 2, pp. 184-193.
	Valentine, S.R., Hanson, S.K. and Fleischman, G.M. (2019), "The presence of ethics codes and employees' internal locus of control, social aversion/malevolence, and ethical judgment of incivility: a study of smaller organizations", <i>Journal of Business Ethics</i> , Vol. 160 No. 3, pp. 657-674.
	van der Kolk, B., van Veen-Dirks, P.M.G. and ter Bogt, H.J. (2019), "The impact of management control on employee motivation and performance in the public sector", <i>European Accounting Review</i> , Vol. 28 No. 5, pp. 901-928.
	Vandenberghe, C., St-Onge, S. and Robineau, E. (2008), "An analysis of the relation between personality and the attractiveness of total rewards components", <i>Relations Industrielles</i> , Vol. 63 No. 3, pp. 425-453.

- Victor, J.A. and Hoole, C. (2021), "Rejuvenating the rewards typology: qualitative insights into reward preferences", *SA Journal of Industrial Psychology*, Vol. 47, pp. 1-17.
- Wainberg, J. and Perreault, S. (2016), "Whistleblowing in audit firms: do explicit protections from retaliation activate implicit threats of reprisal?", *Behavioral Research in Accounting*, Vol. 28 No. 1, pp. 83-93.
- Wen, L., Hao, Q. and Bu, D. (2015), "Understanding the intentions of accounting students in China to pursue certified public accountant designation", *Accounting Education*, Vol. 24 No. 4, pp. 341-359.
- Wen, L., Yang, H.C., Bu, D., Diers, L. and Wang, H. (. (2018), "Public accounting vs private accounting, career choice of accounting students in China", *Journal of Accounting in Emerging Economies*, Vol. 8 No. 1, pp. 124-140.
- Ye, Y. and Lin, L. (2015), "Examining relations between locus of control, loneliness, subjective wellbeing, and preference for online social interaction", *Psychological Reports*, Vol. 116 No. 1, pp. 164-175.
- Zeelenberg, M., Seuntjens, T.G., van de Ven, N. and Breugelmans, S.M. (2020), "When enough is not enough: overearning as a manifestation of dispositional greed", *Personality and Individual Differences*, Vol. 165, p. 110155.

Corresponding author

Sofia M. Lourenço can be contacted at: slourenco@iseg.ulisboa.pt

Variables	Abbreviation	Indicators	Loading
Extrinsic motivation	ExtMot	Because this job has the potential to afford me a certain standard of living Because it allows me to make a lot of money in the future I do this job because it will provide me a good wage in the future Because it provides me with economic certainty	ndix 0.852 0.727 0.728 0.728 0.728
Intrinsic motivation	IntMot	For the status it provides me Because I have fun doing my job	0.701 0.842 0.837
External locus of control	ELC	For the moments or pleasure that this job brings me Many of the unhappy things in people's lives are partly due to bad luck Getting a good job depends mainly on being in the right place at the right time	$0.248 \\ 0.797 \\ 0.101^{a}$
Internal locus of control	ILC .	When I make treet that I have intue inturence over the trungs that happen to me When I make plans, I am almost certain that I can make them work Getting people to do the right things depend upon ability; luck has nothing to do with it	0.698 ^c 0.698 ^c 0.698 ^c
Self-efficacy	SE	What happens to me is my own doing How confident are you that you will be able to achieve most of the goals you have set for yourself? How confident are you that you will accomplish difficult tasks? In general, how confident are you that you can achieve outcomes that are important to you? How confident are you that you will succeed at most any endeavour you set your mind to?	1.00.0 0.871 0.886 0.874 0.874
Preference for financial rewards	PreFinRew	How confident are you that you will successfully overcome many challenges? When things are tough, how confident are you that you can still perform well? Individual performance incentives Basic salary Overall benefits	$\begin{array}{c} 0.881 \\ 0.824 \\ 0.828 \\ 0.634^{c} \\ 0.753 \\ 0.756 \end{array}$
Preference for extrinsic rewards	PreExtRew	Job security Dib security Organization incentive plan Relationships with co-workers Management style Authority/power Workload/job pressure	$\begin{array}{c} 0.437^{b}\\ 0.622^{b}\\ 0.433^{b}\\ 0.443^{b}\\ 0.591^{c}\\ 0.889\\ 0.475^{b}\end{array}$
		Job title/status Fair treatment/equity Team spirit/morale	0.887 0.250 ^a 0.433 ^b <i>continued</i>
Appendix. Variables and indicators' outer loadings		83	Propensity to choose an auditing job

JAOC					
20,6	Loading	$\begin{array}{c} 0.896\\ 0.347^{a}\\ 0.555^{c}\\ 0.553^{b}\\ 0.577^{b}\\ 0.577^{b} \end{array}$	$\begin{array}{c} 0.763\\ 0.763\\ 0.768\\ 0.689^{\rm c}\\ 0.626^{\rm b}\\ 0.641^{\rm c}\\ 0.641^{\rm c}\\ 0.641^{\rm c}\\ 0.647^{\rm b}\\ 0.772\\ 0.686^{\rm c}\\ 0.686^{\rm c}\end{array}$	$\begin{array}{c} 0.885\\ 0.852\\ 0.919\\ 0.902\\ 0.596^{\circ}\end{array}$	a led to an vould lead
84		ort angements nal life	urtunities ies	. myself applying for a job in auditing te a job offer in auditing, I'd accept it o in auditing z a career in auditing ting has a positive connotation	items removed because their value was above 0.4 , but below 0.7 , and removing them ved because, in spite of their value being above 0.4 and below 0.7 , removing them we s preserved in the measurement model are statistically significant at 1% level
	Indicators	Organizational supp Job location Alternative work arr Flexible benefits Balance work-perso Working anvironme	Challenge Job variety Accomplishment Autonomy Responsibility Nature of work Use skill/ability opp Learning opportunit	In a near future, I see If I was invited to ta I'm interested in a jo I see myself pursuin Finding a job in aud	ir value was below 0.4; old values; 'items prese reshold values. All item
	Abbreviation	PreSuppRew	PreIntRew	PAJ	emoved because the rr AVE above thresh CR or AVE below th :s' own creation
Appendix.	Variables	Preference for support rewards	Preference for intrinsic rewards	Propensity to choose an auditing job	Notes: ^a Items r increase of CR o to a decrease of (Source: Author